Author’s response to reviews

Title: Effects of the Proactive Interdisciplinary Self-Management (PRISMA) program on self-reported and clinical outcomes in type 2 diabetes: A pragmatic randomized controlled trial

Authors:

Esther du Pon (estherdupon.hu@gmail.com)
Nanne Kleefstra (kleefstra.langerhans@gmail.com)
Frits Cleveringa (F.G.W.Cleveringa-2@umcutrecht.nl)
Ad van Dooren (ad.a.vandooren@gmail.com)
Eibert Heerdink (rob.heerdink@hu.nl)
Sandra van Dulmen (s.vandulmen@nivel.nl)

Version: 2 Date: 29 Nov 2019

Author’s response to reviews:

Dear Editor,

Thank you for reviewing our manuscript entitled "Effects of the Proactive Interdisciplinary Self-Management (PRISMA) program on self-reported and clinical outcomes in type 2 diabetes: A pragmatic randomized controlled trial" (BEND-D-19-00371)" and considering it for publication in BMC Endocrine Disorders.

A point-by-point reply to your comments can be found on the following page. We trust we completed the necessary revisions to your satisfaction.

We look forward to your response.

Sincerely yours,
on behalf of all co-authors,
1. The corresponding email address provided in the manuscript does not match the corresponding email address provided on the Editorial Manager system. Please amend this.

Yes, the corresponding author has a new email address. We changed it in the manuscript.

2. Please clarify whether the trial was prospectively or retrospectively registered. If it was retrospectively registered please note this in your manuscript alongside information on trial registration.

The trial was prospectively registered.

3. In the "Ethics approval and consent to participate" section, please clarify why formal ethics approval was not necessary for this trial. Please confirm that this complies with national guidelines and provide a reference which supports this.

We added the clarification to the manuscript.

4. Please remove the cover letter, response letter and CONSORT additional files, as they are no longer required at this stage.

Done.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Done.